Author's response to reviews

Title: Genetic Polymorphisms of Nampt Related with the Susceptibility to Esophageal Squamous Cell Carcinoma

Authors:

Zhang Chuanzhen (fax8446@126.com)
Wang Shanshan (951804008@qq.com)
Xu Changqing (xcqys@126.com)
Du Wenjun (crbdwj@126.com)
Ning Tao (ningtao100@126.com)
Liu Changhong (liuchanghong@sdhospital.com.cn)
Zhang Meijuan (zhangmeijuan@sdhospital.com.cn)
Hou Ruiping (houruiping@sdhospital.com.cn)
Chen Ziping (chenziping1966@163.com)

Version: 2 Date: 6 December 2014

Author's response to reviews:

Dear editors:

We have revised the title page, containing the names, institutions, countries and email addresses of all authors. And we added the paper with line and page numbers.

Wish your good news and thank you for your attention and help.

Best Regards.

Ziping Chen

Address: Digestive Department, Shandong Provincial Qianfoshan Hospital, Shandong University, Jingshi Road 16766#, Jinan, 250014, China

E-mail: chenziping1966@163.com

Telephone: +86-0531-82969313

Fax: 0531-82968900